Author's response to reviews

Title: Family Perspectives in Lynch Syndrome; becoming a family at risk, patterns of communication and influence on relations

Authors:

Katarina Bartuma (katarina.bartuma@med.lu.se)
Mef Nilbert (mef.nilbert@med.lu.se)
Christina Carlsson (christina.carlsson@med.lu.se)

Version: 2 Date: 26 April 2012

Author's response to reviews: see over
Dear editor,

Thank You for Your valuable comments on the paper entitled “Family Perspectives in Lynch Syndrome; becoming a family at risk, patterns of communication and influence on relations”.

In response to the following suggestions we have rewritten the discussion and emphasized the subjective and variable responses presented by the informants. Please see the first section of the discussion for the changes made. All authors have taken part in the different steps of the study and have approved the final version and submission. No part of these data have been sent or published elsewhere. None of the authors have any conflicts of interest.

On behalf of the authors,

Katarina Bartuma
MD, PhD